Author's response to reviews

Title: Lessons learned from stakeholders in a facilitation intervention targeting neonatal health in Quang Ninh province, Vietnam

Authors:

Leif Eriksson (leif.eksson@kvh.uu.se)
Duong M Duc (duong_duc2002@yahoo.com)
Ann Catrine Eldh (anncatrine.eldh@ki.se)
Vu Pham N Thanh (thivu@hn.vnn.vn)
Tran Q Huy (huyub@yahoo.com)
Mats Målqvist (mats.malqvist@kvh.uu.se)
Lars Wallin (lars.wallin@du.se)

Version: 2 Date: 16 April 2013

Author's response to reviews: see over
Dear Emily Crow,

Thank you for your question. The Ministry of Health acted as ethical committee in Vietnam. We have revised the manuscript slightly and hope this is sufficient in order for you to proceed with the peer review.

With best wishes

Leif

Leif Eriksson
PhD
International Maternal and Child Health
Uppsala University